Author's response to reviews

Title: Study protocol: Pregnancy and Neonatal Diabetes Outcomes in Remote Australia (PANDORA) Study

Authors:

Louise J Maple-Brown (louise.maple-brown@menzies.edu.au)
Alex Brown (Alex.Brown@sahmri.com)
I-Lynn Lee (i-lynn.lee@menzies.edu.au)
Christine Connors (christine.connors@nt.gov.au)
Jeremy Oats (Jeremy.Oats@thewomens.org.au)
Harold D McIntyre (David.McIntyre@mater.org.au)
Cherie Whitbread (cherie.whitbread@nt.gov.au)
Elizabeth Moore (Liz.Moore@amsant.org.au)
Danielle Longmore (danielle.longmore@menzies.edu.au)
Glynis Dent (glynis.dent@nt.gov.au)
Sumaria Corpus (sumaria.corpus@nt.gov.au)
Marie Kirkwood (marie.kirkwood@menzies.edu.au)
Stacey Svenson (stacey.svenson@bakeridi.edu.au)
Paula van Dokkum (Paula.VanDokkum@bakeridi.edu.au)
Sridhar Chitturi (sridhar.chitturi@nt.gov.au)
Sujatha Thomas (sujatha.thomas@nt.gov.au)
Sandra Eades (sandra.eades@sydney.edu.au)
Monique Stone (Monique.stone09@gmail.com)
Mark Harris (Mark.Harris@mater.org.au)
Chrissie Inglis (manager@healthylivingnt.org.au)
Karen Dempsey (Karen.demspey@nt.gov.au)
Michelle Dowden (michelle.dowden@achwa.org)
Michael Lynch (michael.lynch@nt.gov.au)
Jacqueline Boyle (jacqueline.boyle@monash.edu)
Sue Sayers (Sue.Sayers@menzies.edu.au)
Jonathan Shaw (Jonathan.Shaw@bakeridi.edu.au)
Paul Zimmet (Paul.Zimmet@bakeridi.edu.au)
Kerin O'Dea (Kerin.ODea@unisa.edu.au)

Version: 2 Date: 14 November 2013

Author's response to reviews:

The following has been added to the section titled "Sample Size" on page 14: "The estimated number of participants is at least 700 women."

Additional files have been removed.